Author's response to reviews

Title: Acculturation does not necessarily lead to increased physical activity during leisure time: a cross-sectional study among Turkish young people in the Netherlands

Authors:

Karen Hosper (k.hosper@amc.uva.nl)
Niek S Klazinga (n.s.klazinga@amc.uva.nl)
Karien Stronks (k.stronks@amc.uva.nl)

Version: 3 Date: 2 August 2007

Author's response to reviews: see over
Dear Editors,

We are pleased to hear that our manuscript is suitable for publication in BMC Public Health. As requested we made the following formatting changes to the manuscript:

- Table and figure placement text is removed
- Reference List is renamed into 'References'
- The heading Figures is renamed as 'Figure legends'
- Figures 2-3 are listed separately and the black and white squares are deleted and described with words instead
- Figure title - The titles (Figure 1 etc) within the image files are deleted
- On page 11, ‘figure 2-3’, is changed into figures 2 and 3
- We removed the full stop from the end of the table titles

With regard to the references: two references were double cited in the reference list, these are removed. In the revised version the extracted references match with the cited references in the manuscript (total of 59 references).

Furthermore, the structure of the discussion section is slightly reorganized and small textual changes have been made in order to improve the quality of the paper.

Yours sincerely, also on behalf of the co-authors,

Karen Hosper